Author's response to reviews

Title: Volumetric analysis of central body fat accurately predicts incidence of diabetes and hypertension in adults

Authors:

  Jacob T. Gibby (jacob.gibby@gmail.com)
  Dennis K. Njeru (dennis.njeru@novarad.net)
  Steven T. Cvetko (stevec@novarad.net)
  Ray M. Merrill (ray_merrill@byu.edu)
  Benjamin T. Bikman (benjamin_bikman@byu.edu)
  Wendell A. Gibby (wgibby@novarad.net)

Version: 2 Date: 17 September 2014

Author's response to reviews:

1. Email addresses added for all authors
2. Page and line numbers added
3. Informed consent statement added
4. Competing interests disclosed
5. Authors’ contributions included